Author’s response to reviews

Title: Caught you: Threats to confidentiality due to the public release of large-scale genetic data sets

Authors:

Matthias Wjst (wjst@helmholtz-muenchen.de)

Version: 6 Date: 18 December 2010

Author’s response to reviews:

Formatting changes to be made:
-----------------------------

Major revisions (we require the author to make these changes)

Title: please ensure that the title is identical in the manuscript and the submission system. Currently they do not match.

DONE

References: reference 7 has been duplicated. Please only reference this once and amend the list and main text accordingly.

REMOVED DOUBLE REFERENCE AND REORDERED ALL OTHER

References: Please make sure references are cited in numerical order. We notice reference 21 is not cited within the main text. Please make sure references continue to be cited in numerical order once this reference is cited.

REMOVED

[http://www.dnreunion.org]: Please provide a functioning link to this website.

CORRECTED

Weblinks: Please insert the list of weblinks in the main references list and amend the list accordingly to ensure that all references are presented and cited in consecutive numerical order. Please provide website titles for each weblink.

OK

Minor revisions (we can make these changes for you, although it will speed up publication of your manuscript if you do them while making the major changes above)

Abstract: Please ensure that the abstract in the submission system contains the headings 'Background, Discussion, Summary'.

DONE
Abbreviations: Please format these in to a paragraph.

DONE

References: please provide the full citation details for references 3, 7 and 18. You can view the BMC reference guide at the following link http://www.biomedcentral.com/bmcgenet/ifora/#references.

NOT SURE WHAT'S WRONG WITH REFERENCE 3 - THE BOOK IS HERE http://books.google.de/books?hl=de&lr=&id=kn7OBM3J1qQC&oi=fnd&pg=PR9&dq=lawrie+genetic+privacy&sig=c4LtYiYN9LIc45lVwswMGyyyZm8#v=onepage&q&f=false

References: the reference list should contain up to 30 authors' names for each citation. The term 'et al.' should not be used if there are fewer than 30 authors.

OK

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.

OK - WOULD LIKE TO SEE ALSO THE ONLINE PROOF AS DURING THE LAST BMC EDITING THERE HAD BEEN INTRODUCED SEVERAL ERRORS THAT WERE EXTREMELY DIFFICULT TO CORRECT LATER.

THANKS FOR ALL YOUR WORK!